% !TEX encoding = UTF-8 Unicode
% !TEX TS-program = pdflatex
% !BIB TS-program = biber
\documentclass{article}
\usepackage[T1]{fontenc}
\usepackage[utf8]{inputenc}

\usepackage[style=italian]{csquotes}
\usepackage[%
backend=biber,
style=philosophy-modern,
annotation=true,
hyperref,
backref]{biblatex}
\addbibresource{philosophy-examples.bib}

\usepackage{hyperref}

\newcommand{\cmd}[1]{\texttt{\textbackslash #1}}

\begin{document}

\textbf{N.B.} For other examples see \texttt{philosophy-examples.bib} and \texttt{biblatex-examples.bib} files.

\section{Standard commands}

\begin{description}
\item[\cmd{cite\{eco:2004\}}] \cite{eco:2004}\\
\item[\cmd{textcite\{eco:2004\}}] \textcite{eco:2004}\\
\item[\cmd{parencite\{eco:2004\}}] \parencite{eco:2004}\\
\item[\cmd{autocite\{eco:2004\}}]\autocite{eco:2004}
\end{description}

\section{Special commands}

\subsection{\cmd{sdcite\{key\}}}

This command is deprecated. Use the \texttt{entrysubtype} mechanism  instead: \cite{aristotle:physics}  (see documentation for details).

\begin{description}
\item[\cmd{sdcite\{guzman:sd\}}] \sdcite{guzman:sd}
\end{description}

\subsection{\cmd{footcitet}}

Typeset a footnote using the same style of \cmd{textcite}.\footcitet[12-13]{eco:2004} Compare with this one.\footcite[12-13]{eco:2004}

\section{Some miscellaneous citations}

\textcite{Agati1991,Alpers1991,Mori:2008,comenio:oo,eco:2004,heidegger:sz,pantieri:artelatex,Poincare:1968-ORIG,popper-logik,descartes:discours,kant:kpv:xref,kant:ku:xref,Spencer:1891vol2,guzman:sd,Berti:1977,Spencer:1891b,berto:sep-impossible-worlds,corrocher:2009,federspil:2009,Valbusa:2007,britannica,ctan,guit:sito,SEP}

\printbibliography

\printbibliography[restoreclassic,type=book]

\end{document}
